Author's response to reviews

Title: Fostering implementation of health services research findings into practice: A consolidated framework for advancing implementation science

Authors:

- Laura J Damschroder (laura.damschroder@va.gov)
- David C Aron (david.aron@va.gov)
- Rosalind E Keith (rekeith@umich.edu)
- Susan R Kirsh (susan.kirsh@va.gov)
- Jeff A Alexander (jalexand@umich.edu)
- Julie C Lowery (julie.lowery@va.gov)

Version: 8 Date: 28 July 2009

Author's response to reviews:

Hello -

I have "accepted" all revisions, as requested.

The abstract was included in the previous version on the 2nd page. I've labeled it as such in this version.

Please let me know if there are any further issues or questions.

Best Regards -
Laura